Author’s response to reviews

Title: Prevalence and associated risk factors of human intestinal parasitic infections: a population-based study in the southeast of Kerman province, southeastern Iran

Authors:
Mohammad Javad Abbaszadeh Afshar (Mj.afshar1990@gmail.com)
Maryam Barkhori Mehni (m.bmehni1980@gmail.com)
Mostafa Rezaeian (rezaiian@tums.ac.ir)
Mehdi Mohebali (mohebali@tums.ac.ir)
Vali Baigi (vbaygi@yahoo.com)
Somayeh Amiri (Mj.afshar1990@yahoo.com)
Mohammad Bagher Amirshekari (Mj.afshar1990@outlook.com)
Ruhollah Hamidinia (toorajharati@gmail.com)
Mohammad Samimi (danyalimani70@gmail.com)

Version: 3 Date: 18 Dec 2019

Author’s response to reviews:

December 18, 2019

BMC Infectious Diseases

Dear editor

Thank you very much for reviewing our manuscript. Please find attached a point-by-point response to the Editor and Assistant Editor comments with this cover letter. We hope that you find our responses satisfactory.

Sincerely

Maryam Barkhori
Vice-President of Health Affairs, Jiroft University of Medical Sciences, Jiroft, Iran
E-mail: m.bmehni1980@gmail.com
Tel: +989133486284

Mohammad Javad Abbaszadeh Afshar
1. Please include an “Abbreviations” section after the “Conclusions” section of your manuscript where you list all of the abbreviations used in the manuscript. All abbreviations should still be defined in the text at first use. 
The “Abbreviations” section added after the “Conclusions” section.

2. In the section 'Funding', please also describe the role of the funding body/bodies in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript. 
The role of the Jiroft University of Medical Sciences as funding body describe in the “Funding” section.

3. Please provide a 'Declarations' heading for your declarations, below the conclusion. 
Done

4. Please rename 'Declaration of Competing interest' to 'Competing interests'.
Done

5. In the section 'Ethics Approval and Consent to Participate', please state whether the informed consent obtained was written or verbal. If verbal, please state the reason and whether the ethics committee approved this procedure. 
In this study, an informed written consent was obtained from the participants or their parents, in term of children.

6. Please state in the cover letter whether the map depicted in figure 1 is your own or taken from another source. If taken from another source please acknowledge the source in the figure legend, and if it is under copyright also state the written permission given to use and adapt it. 
The map depicted in figure 1 is prepared by the authors of this study and adjusted for the size by GIS software (ArcMap).

7. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.

A single and clean version of the manuscript submitted. The figure (1) and Tables (1-3) submitted as separate files.